Author's response to reviews

Title: Rare ileal localization of angiolipoma presenting as chronic haemorrhage and severe anemia: Case report

Authors:

Nicola Della Volpe (nicola.dellavolpe@unina.it)
Luigi Bianco (l.bianco7@alice.it)
Claudio Bonuso (clbonuso@tin.it)
Mario Annecchiarico (mannecchiarico@hotmail.com)
Raffaele D'Avino (ramada1@libero.it)
Pierino Di Silverio (pierodisilverio@libero.it)
Maria Rega (mariarega@virgilio.it)
Assunta Caiazza (assunta.caiazza@unina.it)
Maria D'Armiento (madarmie@unina.it)
Graziano Pernazza (gpernazza@libero.it)

Version: 2 Date: 5 August 2007

Author's response to reviews: see over
Dear Editor,

I re-submitted you this manuscript on interesting and rare case of Angiolipoma of Ileum. It seems to be the first case in the literature (in all there are only two cases of angiolipoma of ileum) presenting without a painful specific symptomatology or an alvus alteration. I hope you give positive consensus to it to publish on your journal.

Thank you.
Claudio Bonuso, MD
Department of General and Geriatric Surgery, Oncological and Advanced Technologies. University of Naples "Federicoll", Italy